Author’s response to reviews

Title: Identification of a potent herbal molecule for the treatment of breast cancer

Authors:

Srinivas Koduru (srinivas.koduru@uky.edu)
Sowmyalakshmi Srinivasan (ssrin3@uky.edu)
Raj Kumar (rkuma4@uky.edu)
Jürgen Rohr (jrohr@uky.edu)
Chendil Damodaran (dchen2@uky.edu)
Rohini Gomathinayagam (rgoma4@uky.edu)

Version: 4 Date: 3 December 2008

Author’s response to reviews:

Dear Editor,

We are resubmitting our final revised manuscript entitled “Identification of a potent herbal molecule for the treatment of breast cancer (MS 1296378060212447)” after making the necessary modifications as per the editors’ comments for your kind perusal. The following modifications have been made:

• Background section: highlighting area has been removed in Paragraph 1.

• Figure numbers are removed from the figures and table has been made using table tools and included in text after references.

Please do not hesitate to contact me if you have any question.

Thanking You

Sincerely yours

Damodaran Chendil, PhD